Author’s response to reviews

Title: Effective behavioral intervention strategies using mobile health applications for chronic disease management: a systematic review

Authors:

Jung-Ah Lee (jungahl@uci.edu)

Mona Choi (monacho@yuhs.ac)

Sang A Lee (m1sslee1011@gmail.com)

Natalie Jiang (nmjiang@uci.edu)

Version: 2 Date: 23 Dec 2017

Author’s response to reviews:

Dear Reviewers,

Thank you for the opportunity to revise our manuscript, entitled “Effective behavioral intervention strategies using mobile health applications for chronic disease management: a systematic review” for publication in BMC Medical Informatics and Decision Making, based on the thoughtful review we received. We believe we have addressed each point from the review in the revised manuscript. Below, we review each point in detail and note how and where we have addressed each concern in our revised manuscript. We also highlight the changes in red in our revision.

Thank you.